Author's response to reviews

Title: Epidemiological and clinical characteristics of childhood pandemic 2009 H1N1 virus infection: An observational study

Authors:

Jung-Woo Rhim (benign7@hamail.net)
Kyung-Yil Lee (leekyungyil@catholic.ac.kr)
You-Sook Youn (vanco77@hanmail.net)
Jin-Han Kang (kjhan@catholic.ac.kr)
Ji-Chang Kim (jichang71@hotmail.com)

Version: 2 Date: 9 May 2011

Author's response to reviews: see over
Dear Dr. Patel and Editors

We are pleased to enclose this manuscript for consideration for publication in *BMC infectious diseases* in the section ‘Research article’.

We revised some errors in numbering of figures (Fig. 3-5) in Figure legends (page 19) of the first manuscript. We also removed the additional files which were our published articles.

We thank your consideration for our article and look forward to hearing from you.

Kind regards
Kyung-Yil Lee, MD, on behalf of the authors
Department of Pediatrics, The Catholic University of Korea,
Daejeon St. Mary's Hospital, 520-2 Daeheung-dong, Jung-gu,
Daejeon 301-723, Republic of Korea
Tel; +82-42-220-9541, Fax; +82-42-221-2925, E-mail; leekyungyil@catholic.ac.kr